Author's response to reviews

Title: Psychometric Properties of the Chinese Version of the Fatigue Scale-Adolescent

Authors:

William H.C. Li (william3@hku.hk)
Ka Yan Ho (eva19841226@yahoo.com.hk)
Ka Wai Katherine Lam (kathins@gmail.com)
S.Y. Chui (dieselfrankie@yahoo.com.hk)
Chi-Fung Godfrey Chan (devilbb@hku.hk)

Version: 4 Date: 2 April 2015

Author's response to reviews:

Dear Editor,

Please note the following changes:

(1) Line numbering: has been re-formated as continuously numbered and not per page

(2) Figure 1 has been re-uploaded

Thanks,

Authors